Author's response to reviews

Title: Evaluation of Noise Hazard During the Holmium Laser Enucleation Of Prostate

Authors:
Huan Xu (xuhuaneric@hotmail.com)
Yanbo Chen (chenyanbo126@126.com)
Meng Gu (gumengGM@126.com)
Qi Chen (qiqi_chenqi@yeah.net)
Zhong Wang (zhongwang2000@sina.com)

Version: 2 Date: 29 Jun 2017

Author’s response to reviews:

MS No. BURO-D-16-00168R1

Title: Evaluation of Noise Hazard During the Holmium Laser Enucleation Of Prostate
Authors: Huan Xu, Yan-bo Chen, Meng Gu, Qi Chen, Zhong Wang

Dear Editor:

Thank you for your review of our manuscript. My manuscript, referenced above, has been revised according to the reviewers’ and editors’ comments. The modifications are as follows:

In your manuscript you state that "43 of the 59 questionnaires are considered valid;". Please revise your manuscript to clarify what you mean by this statement and to explain why 16 questionnaires were excluded from your study.

Thank you very much for your remind. We excluded that 16 questionnaires which were incomplete or with logic mistakes. This has been added to the manuscript.

We recommend that you copyedit the paper to improve the style of written English.

We have send it to the one who is fluent in English and with his help this manuscript is revised.

Please revise your ethics statement (declarations section) to provide the name of the ethics committee that approved your study.
Ethical approval is by Ethics Committee in the 9th people's hospital in Shanghai. We have added this to the manuscript. This has been added in the manuscript.

Please revise your Consent for publication statement to clarify that 'consent for publication' was not required for this study.

We have added this in the last of the test.

We would like to thank all of the reviewers and editors for introducing the literature to us. This is very helpful to our future study. We have confirmed that all author details on the revised version are correct, that all authors have agreed to authorship and order of authorship for this manuscript and that all authors have the appropriate permissions and rights to the reported data. If there are other errors or further requests, please contact us by e-mail.

Yours sincerely,

Huan Xu

Email: xuhuaneric@hotmail.com

Corresponding author: Zhong Wang
Department of Urology, Shanghai Ninth People’s Hospital, Shanghai Jiaotong University School of Medicine, Shanghai 200011, China
Email: zhongwang2000@sina.com